Author's response to reviews

Title: CMV infection of liver transplant recipients: comparison of antigenemia and molecular biology assays

Authors:

   Maria L Amorim (mlamorim@net.sapo.pt)
   Jose M Cabeda (jcabeda@mail.telepac.pt)
   Rui Seca (rseca@isoterica.pt)
   Ana C Mendes (aconstanca@sapo.pt)
   Ana P Castro (jrcastro@ntc.pt)
   Jose M Amorim (microbiologia@clix.pt)

Version: 6 Date: 22 May 2001

changes done according to items 1-5 of Dr. Carole letter